Author's response to reviews

Title: Down-regulation of the expression of CCAAT/enhancer binding protein alpha gene in cervical squamous cell carcinoma

Authors:

Zemin Pan (panteacher89@sina.com)
Weinan Zheng (rwz77666@163.com)
Jinli Zhang (zhangjinli@shzu.edu.cn)
Rui Gao (gaorui_shz@sina.com)
Dongmei Li (lidong_abc@126.com)
Xiaoqing Guo (gxqxj66@sina.com)
Hu Han (49166070@qq.com)
Feng Li (lifeng7855@yahoo.com.cn)
Shen Qu (qushen@mails.tjmu.edu.cn)
Renfu Shao (rshao@usc.edu.au)

Version: 6 Date: 22 April 2014

Author's response to reviews: see over
EDITORIAL COMMENT:

**Thank you for your revision, however we have noticed that you state in your description of figure 4a and figure 4b that the statistical significance was (P=0.000). Please could you clarify whether this is the case or whether it is a typographical error.

Answer:

In the results of this manuscript with description for figure 4a and figure 4b are statistical significance (P=0.000). This is typographical error, it should write for figure 4c with statistical significance. We have correct this error on the description of results and revised on figure legends 4.